Bayou Corne Checklist

Name: Daryl Williams
Date: 7/12/2013
Napoleonville Field (7010) -- Assumption Parish, LA

Shut-In/Flowing | Thg Pressure (psig) Choke (/64") Notes
ORW #1 Shut-In 28
ORW #2 Shut-In <2
ORW #3 Shut-In N/A No gas
ORW #4 Flowing 44 2
ORW #5 N/A N/A Inaccessible
ORW #6 Flowing 38 2
ORW #7 N/A N/A Inaccessible
ORW #8 N/A N/A Inaccessible
ORW #9 Shut-In 52
ORW #10 Flowing 55 3
ORW #11 Shut-In 7
ORW #12 Flowing 10 1 Possible intent to P&A as early as next week
ORW #13 Shut-In 36
ORW #14 Flowing 46 6
ORW #15 Flowing 21 5
ORW #16 Shut-In 23
ORW #17 Shut-In 5.5
ORW #18 Shut-In 51
ORW #19 Flowing 56 1
ORW #20 PROPOSED PROPOSED PROPOSED Proposed Well
ORW #21 Shut-In 0 -
ORW #22 Flowing 14 1
ORW #23 Shut-In 46
ORW #24 Flowing 56 1
ORW #26 Shut-In 18
ORW #27 Shut-In 0 No gas
ORW #28 Shut-In 17
ORW #29 Shut-In 44
ORW #30 Shut-In 55
ORW #31 Shut-In 30
ORW #32 Shut-In 22
ORW #33 Shut-In 0 No gas
ORW #36 Flowing 49 5




Shut-In/Flowing | Tbg Pressure (psig) Choke (/64") Notes
ORW #37 Flowing 37 1 Grass needs to be cut
ORW #38R Shut-In 36 In process of dewatering, 11am
ORW #39 Flowing 46 1.5
ORW #40 Flowing 44 1 Pumped water the day before
OGRW#1 Flowing 58 4
Shut-Tn/Flowing
(if applicable) Tbg Pressure (psig) | Csg Pressure (psig) Notes
G-1 Shut-In - 215
OG #3-A Shut-In 590 598 Tagging: 10am, will run sonar on cavern and run psi,temp,gamma grad log
OG #1 Shut-In 118.9 245.6
TBC #1 Shut-In 25 215 Possible intent to P&A as early as next week
TBC #2 Shut-In 56 - Possible intent to P&A as early as next week
TBC #3 Shut-In 0 0
BC-2 Shut-In 54 21 Grass Needs to be cut
OG Water Well #3 Sinkhole
Flowing or Locked-Out? Cumulative Vegetation Hauled Off:
loads of 25-yd roll off boxes
Locked-Out Cumulative Oil Hauled Off:
bbls
Notes: Notes:|Skimming, and excavator on location sorting thru debris. Didn't
see any trucks removing debris offsite. More pronounced
hydrocarbon scent than my previous visit a week ago.
Additional Remarks

Working on pad for MMRA-6 near Texas Brine facility. Building a slip ROW back behind the ORW 13 for continued CPT analysis via a barge or other
swamp vehicle. The were running a line to tag the cavern via the 3-A as of 10am. Scheduled to run a sonar and a pressure, temperature, gamma
gradient log afterwards. There was heavy rain late yesterday over the area. The north and west berms looked good. All are up to grade. Was asked to
remain off the berms so that TxBrine could conserve the berm with intent in the near future to lay limestone or other rock over the top to make it

permanent. Current rock across northern berm road to 39 and 40 was very good and drivable. Area around 4 and 11 was not and was accessed via ATV.
Road to 5,7 and 8 between the berms was soft so | remained off of it. North berm and west berm to the ORW 9 looked good but | remained off as well
as it had not been laid with rock as of yet. | accessed the 9 via the 16 opposite Gumbo Street. Board has been laid on upslope from access road at the 9
to the berm but the drop off is too steep for a car. Truck and ATV, ok. Was told that there may be intent to plug the wells on Well Pad 3 as early as next
week but it is not confirmed. This may include the TBC 1 & 2, and the ORW 12, not confirmed. The water well may remain.




